Author’s response to reviews

Title: Remodeling of extracellular matrix by normal and tumor-associated fibroblasts promotes cervical cancer progression

Authors:

Alexandra Fullár (fullarsz@gmail.com)
József Dudás (jozsef.dudas@i-med.ac.at)
Lászlóné Oláh (julcsi@korb1.sote.hu)
Péter Hollósi (phollosi@yahoo.com)
Zoltán Papp (maternity@maternity.hu)
Gábor Sobel (gabor@sobel.hu)
Katalin Karászi (katalin.karaszi@nds.ox.ac.uk)
Sándor Paku (paku@korb1.sote.hu)
Kornélia Baghy (kori@korb1.sote.hu)
Ilona Kovalszky (koval@korb1.sote.hu)

Version: 5 Date: 19 March 2015

Author’s response to reviews: see over
Dear Professor Solera,

We are pleased to understand that our manuscript is considered for publication in the BMC Cancer Journal. Please find attached our revised manuscript entitled “Remodeling of extracellular matrix by normal and tumor-associated fibroblasts promotes cervical cancer progression”. We carried out the last modifications of the manuscript as it is listed below.

Responses to Editorial’s request

1. Format Authors’ contribution according to our guidelines.
   The Authors’ contribution section has been formatted.
2. Re-name Funding section as Acknowledgement.
   The Funding section has been renamed to Acknowledgement.
3. Include tables at the end of the main ms and remove from additional files.
   Tables have been included at the end of the manuscript and been removed from additional files.

Corrected parts of the manuscript are highlighted in blue colour.
We do hope, that the changes meet your requirements.

Best regards,

Ilona Kovalszky  M.D, Ph.D, D.Sc
Professor of Pathology